DISCUSSION PAPER SERIES

No. 9157

ENTREPRENEURIAL INNOVATIONS AND TAXATION

Andreas Haufler, Pehr-Johan Norbäck and Lars Persson

INDUSTRIAL ORGANIZATION



Centre for Economic Policy Research

www.cepr.org

Available online at:

www.cepr.org/pubs/dps/DP9157.asp

ENTREPRENEURIAL INNOVATIONS AND TAXATION

Andreas Haufler, University of Munich Pehr-Johan Norbäck, Research Institute of Industrial Economics Lars Persson, Research Institute of Industrial Economics and CEPR

> Discussion Paper No. 9157 September 2012

Centre for Economic Policy Research 77 Bastwick Street, London EC1V 3PZ, UK Tel: (44 20) 7183 8801, Fax: (44 20) 7183 8820 Email: cepr@cepr.org, Website: www.cepr.org

This Discussion Paper is issued under the auspices of the Centre's research programme in **INDUSTRIAL ORGANIZATION**. Any opinions expressed here are those of the author(s) and not those of the Centre for Economic Policy Research. Research disseminated by CEPR may include views on policy, but the Centre itself takes no institutional policy positions.

The Centre for Economic Policy Research was established in 1983 as an educational charity, to promote independent analysis and public discussion of open economies and the relations among them. It is pluralist and non-partisan, bringing economic research to bear on the analysis of medium- and long-run policy questions.

These Discussion Papers often represent preliminary or incomplete work, circulated to encourage discussion and comment. Citation and use of such a paper should take account of its provisional character.

Copyright: Andreas Haufler, Pehr-Johan Norbäck and Lars Persson

CEPR Discussion Paper No. 9157

September 2012

ABSTRACT

Entrepreneurial innovations and taxation*

Stimulating entrepreneurship is high on the policy agenda of many countries. We study the effects of tax policies on entrepreneurs' choice of riskiness (quality) of an innovation project, and on their mode of commercializing the innovation (market entry versus sale). Limited loss offset provisions in the tax system induce entrepreneurs innovating for entry to choose projects with inefficiently little risk, whereas this imperfection does not arise when innovating for sale. Tax systems which systematically favor market entry of entrepreneurs can thus lead to welfare losses due to inefficient quality choices, despite leading to more competition in the product market.

JEL Classification: H25, L13, M13 and O31 Keywords: business taxation, innovation and market entry

Andreas Haufler Seminar for Economic Policy Akademiestr. 1/ II D-80799 Munich Germany	Pehr-Johan Norbäck The Research Institute of Industrial Economics (IUI) Box 5501 S-11485 Stockholm SWEDEN
Email: haufler@irz.uni-muenchen.de	Email: PJN@iui.se
For further Discussion Papers by this author see: www.cepr.org/pubs/new-dps/dplist.asp?authorid=110767	For further Discussion Papers by this author see: www.cepr.org/pubs/new-dps/dplist.asp?authorid=153438

Lars Persson Research Institute of Industrial Economics (IFN) Box 55665 S-102 15 Stockholm SWEDEN

Email: lars.persson@ifn.se

For further Discussion Papers by this author see: www.cepr.org/pubs/new-dps/dplist.asp?authorid=134788

*Paper presented at conferences and seminars in Ann Arbor, Berlin, Frankfurt, Lund, Milan, Munich, Oxford and Stockholm. We have benefited from many helpful comments and suggestions by Massimo Bordignon, Michael Devereux, Judith Freeman, Clemens Fuest, Volker Grossmann, Dietmar Harhoff, Magnus Henrekson, Christian Keuschnigg, Andreas Oestreicher, Marco da Rin, Martin Ruf, Mikael Stenkula and Miranda Stewart. Thanks also to Christina Lönnblad for improving the language and to Nina Öhrn for research assistance. Financial support from the Jan Wallander and Tom Hedelius' Research Foundations is gratefully acknowledged. This paper was started when the first author visited the Research Institute of Industrial Economics in Stockholm. He wishes to thank the members of the Institute for their hospitality.

Submitted 24 September 2012

1 Introduction

How to stimulate entrepreneurship is one of the pressing policy challenges facing many countries in the world today. In Europe, the European Commission (2008) launched the "Small Business Act for Europe" in June 2008, in response to this challenge. This document sets out a comprehensive policy framework for the EU and its member states to create an environment that rewards entrepreneurship, specifically mentioning taxation in this context. At the same time, several EU countries are reforming their tax systems to make them more compatible with entrepreneurship. In the Netherlands, for example, the 2011 Tax Plan comprises a whole set of measures explicitly aimed at encouraging entrepreneurship. Another example is Sweden, which is currently making the largest investigation of its corporate tax system in 20 years, with the main objective of making the tax system more conducive to entrepreneurship.¹

It is well known that taxation will reduce the amount of entrepreneurial activities.² However, little is known about how the tax system will affect the qualitative (i.e., risk-related) dimension of entrepreneurship, which might be as important for growth and welfare. Cohen (2010, p. 137) who reviews the empirical literature on firms size and R&D, concludes that small firms tend to account for a disproportionately large share of innovations, and that - where data exist - they also tend to pursue more significant innovations than incumbents.³ The purpose of this paper is to increase our knowledge about how the tax system affects this qualitative dimension of entrepreneurship.

When evaluating the effects of taxes on entrepreneurial decisions, it is important to take into account that entrepreneurs not only commercialize their inventions or business ideas by entering the product market, but also by selling them to incumbent firms. Indeed, a substantial share of inventions made by independent innovators is commercialized through the sale or the licensing of a patent.⁴ The question of how the tax

¹One of the proposed measures is to allow start-up firms to delay the payment of social security fees for several years. See the report "Tax incentives for venture capital", http://regeringen.se/sb/d/15680/a/185102 (in Swedish).

²See Henrekson and Sanandaji (2011) for a recent survey of this literature.

³This conclusion is confirmed by Henkel et al. (2010) in a recent empirical study of the electronic design automation industry. The importance of small entrepreneurial firms in the innovation process in the United States is documented in Baumol (2004).

⁴Serrano (2010, Table 1) reports for the United States that entrepreneurs sold 17.5% of their patents during the period 1983-2001, and this share increases to 24% if the patents are quality-weighted with

system affects the entrepreneur's interdependent choices of an R&D project and the mode of commercializing the innovation is therefore at the center of our analysis.

We focus on two fundamental features of existing tax systems. First, many countries effectively tax-favor the market entry of entrepreneurs over the alternative of selling a business or patent. In this regard, it is important to recognize that the sale of a business is classified as a capital gain by some countries, but as ordinary personal business income by others. As Table 1 shows, the majority of countries in continental Europe taxes the sale of a business as ordinary income, and at tax rates that are substantially above the corporation tax (which applies to most entrepreneurial businesses). Another source of a tax advantage of market entry is that the income tax that is due under project sale incorporates the value of the entire firm or patent, whereas the profit tax paid under market entry is based only on current-period profits. Finally, several countries support small, incorporated businesses by means of reduced corporate tax rates, as well as other tax breaks.⁵ Together, these provisions imply that many countries in Europe and elsewhere grant tax advantages to entrepreneurs who market their inventions themselves, rather than selling out to an incumbent firm.

At the same time, tax systems in all OECD countries incorporate a fundamental asymmetry as positive earnings are taxed immediately, whereas losses can only be offset against positive incomes.⁶ The rationale for this asymmetry, which we take as given in our analysis, is to prevent tax fraud. Small entrepreneurs, who are developing a single business idea, will therefore not be able claim a tax rebate when they invest in a project that turns out to be unsuccessful. In Germany, for example, restrictive loss offset provisions for entrepreneurs are regarded as one of the main obstacles for the development and growth of small, innovative businesses (EFI, 2011, p. 19). In contrast, restrictions on loss offset opportunities are less important when the project is sold to a large, incumbent firm, as the incumbent is more diversified and will likely be able to offset losses on a new investment against positive taxable profits earned in its other

the number of citations received. Furthermore, a large-scale survey carried out in six EU countries suggests that roughly 10% of all patents are licensed (Giuri et al., 2007).

⁵See, e.g., the chapter on small business taxation in Mirrlees at al. (2011), which describes in detail the numerous tax privileges enjoyed by small, incorporated businesses in the United Kingdom.

 $^{^{6}}$ In contrast, Bond and Devereux (2003) stress that a business tax system that is neutral in the presence of uncertainty would require the government to pay a tax rebate equal to the full value of the initial investment in the case of default.

	classification of	tax rate for	corporate	
country	business sale a	business sale (%) b	tax rate (%) c	
Austria	capital gain	25	25	
Belgium	capital gain	$16.5/33 {}^{d}$	34	
Denmark	ordinary income	44.3	25	
Finland	ordinary income	49	26	
France	capital gain	29	34.4	
Germany	ordinary income	45	30.2	
Ireland	capital gain	25	12.5	
Italy	ordinary income	44.9	27.5	
Netherlands	ordinary income	52	25	
Norway	ordinary income	40	28	
Spain	ordinary income	43	30	
Sweden	ordinary income	$26.3/56.5 \ ^{e}$	26.3	
United Kingdom	capital gain	10/28 f	28	
United States	capital gain	15	39.2	

Table 1: Taxation of business sale vs. corporate tax rate (2011)

^a including taxation of patent sales and royalties

 b for capital gains: capital gains tax rate; for ordinary income: top national and representative sub-central tax rate, not including social security contributions

 c regular central and sub-central corporate tax rate

 d lower rate applicable to sale of business; higher rate applicable for sale of patents

 e lower rate applicable to business income of individuals that is retained in a reserve fund

 f lower rate applicable to the first GBP 5 million from sale of a business

<u>Sources:</u> European Tax Handbook (2011). International Bureau of Fiscal Documentation, Amsterdam. OECD Tax Database: www.oecd.org/ctp/taxdatabase operations (see Mirrlees et al., 2011, p. 454).⁷

Against this background we develop a model where a European-style tax system influences the decisions of the entrepreneur to select an R&D project, and to choose the mode of commercializing the innovation. In the initial stage, the entrepreneur chooses the quality of an innovation, trading off higher returns against an increased risk of failure. Before the success of the project is known, the entrepreneur can either sell the project to an incumbent, or keep it. A characteristic element of our model is that the equilibrium sales price of the firm or patent is explicitly derived from the valuation of the risky innovation by both the incumbents and the entrepreneur. These valuations are affected by the tax considerations described above. In particular, the entrepreneur will benefit from lower taxes under market entry, but only if the project is successful.

We first establish that whenever the entrepreneur enters the market in equilibrium, she will choose an innovation that involves too little risk, in comparison to the socially desirable project. This distortion arises because there is an extra cost of failing in terms of lost tax deductions, and hence the entrepreneur finds it optimal to reduce the risk of being left with non-deductible investment outlays. In contrast, when innovating for sale the entrepreneur will sell her invention to an incumbent before all uncertainty is revealed. This guarantees a full deduction of her investment costs, whereas competitive bidding by incumbents ensures that the entrepreneur receives the true market value of the innovation. As a result, the entrepreneur will choose a project with efficient risk characteristics when innovating for sale.

Based on these results, our analysis evaluates the consumer welfare and employment effects of tax policies towards entrepreneurship. One of the main policy objectives of a favorable tax treatment of new, small firms is the belief that they create additional jobs.⁸ This objective is particularly relevant in countries suffering from inefficient labour

⁷There is substantial empirical evidence for the importance of this fundamental asymmetry in the corporate tax code. Auerbach (2007, Table 4) documents for the United States that the aggregate losses reported by corporations rose to 350-400 billion USD per annum in each of the years 2001-2003 (representing roughly two thirds of positive corporate profits in the same years), a large part of which was not tax deductible for the firms that incurred the losses. Econometric studies have furthermore shown that the asymmetric tax treatment of profits and losses has significant effects on entrepreneurial risk-taking (Cullen and Gordon, 2007), as well as on firms' investment incentives (Edgerton, 2010).

⁸A substantial part of the net employment growth in the SME sector comes from a small number of high-growth firms, so-called 'Gazelles', which are typically young and are represented in all industries (Henrekson and Johansson, 2010).

markets and high unemployment. We show that tax policies to support entrepreneurial market entry can indeed lead to increased production and increased labour demand in concentrated markets. From a welfare perspective, however, the optimal policy will involve an equilibrium with project sale for a wide range of parameter values, as this is associated with an efficient R&D project choice. In a model extension where innovations reduce variable production costs, rather than only fixed costs, it is even possible that policies to promote small businesses will be directly counterproductive by reducing *both* consumer surplus and total employment in the industry.

These results imply that an isolated reduction in corporate taxes that promotes entrepreneurial market entry can be counterproductive, even if it increases the incentives for entrepreneurship. These findings are in line with recent empirical results showing that lower corporate taxes induce entry by new firms (Djankov et al., 2010; Da Rin et al., 2011), but also reduce the size and the capital intensity of the new entrants and thus lead to 'weaker' firms, on average (Da Rin et al., 2010). Our analysis adds a qualitative dimension to this picture, by focusing on the endogenous choice of entrepreneurs' R&D projects. We thus conclude that a reduction in corporate tax rates might need to be accompanied by a reduction in the taxes associated with selling business, in order to spur high-risk, but also high-return innovations.

Our model brings together two different strands in the literature. On the one hand is the public finance literature that analyzes the effects of taxes on various decision margins of entrepreneurs. Several contributions focus on the progressiveness of the personal income tax schedule as an obstacle to entrepreneurial activity (e.g. Gentry and Hubbard, 2000). In contrast, Gordon (1998) stresses that start-up enterprises have the option of incorporating, thus benefiting from the widespread fall in corporate tax rates over the last few decades. Fuest et al. (2002) show that the presence of a substantial positive tax gap between the personal income tax and the corporation tax is a second-best solution to the problem of asymmetric information faced by new firms. Keuschnigg and Nielsen (2002, 2004) focus on the effects of various tax policies when entrepreneurs face financial constraints and set up a contract with a venture capitalist under conditions of one-sided or two-sided moral hazard. None of these contributions, however, incorporates a choice between different R&D projects, nor the option for the entrepreneur to sell her invention to an incumbent firm.⁹

⁹Among the empirical contributions, Cullen and Gordon (2007) estimate the effects of imperfect loss offset provisions and of rate differences in personal and corporate income taxation on entrepre-

Secondly, this paper is also related to the literature on R&D and market structure, which mainly focuses on the level of R&D efforts.¹⁰ Several papers study the type of R&D project undertaken by firms and entrepreneurs (e.g. Bhattacharya and Mookherjee, 1986). There is also a literature on entrepreneurship and innovations, which is summarized in Acs and Audretsch (2005). To our knowledge, the only analysis considering how the entry mode affects the type of R&D is Färnstrand Damsgaard et al. (2010). However, this paper focuses on the interaction between entrepreneurial and incumbent innovations and abstracts from tax policies, which are central to the present study.¹¹

The remainder of the paper is organized as follows. Section 2 outlines our benchmark model, where the innovation reduces only fixed costs. In Section 3, we solve the different stages of the model and determine the equilibrium allocation in different tax regimes. Section 4 analyzes the effects of tax policy on the equilibrium project choice and commercialization mode, as well as on welfare and employment in the industry. Section 5 analyzes the case where the innovation reduces variable costs of production. Section 6 discusses several other model extensions and Section 7 concludes.

2 The framework

We consider an imperfectly competitive market with n identical incumbent firms. Entry costs deter further firms from entering the market, unless they have a superior technology. The focus of our analysis lies on the decisions of an independent innovator, or entrepreneur, who chooses a project with certain risk characteristics and decides whether to sell the invention or try to enter the market herself. In line with the evidence

neurial risk taking. Egger et al. (2009) analyze the incorporation decision of entrepreneurs and provide empirical evidence that a positive tax gap between personal and corporate tax rates favors incorporation. Bloom et al. (2002) and Ernst and Spengel (2011) show that R&D tax credits have positive and significant effects on the volume of R&D, but do not address the choice of quality (or riskiness) of an innovation project.

 $^{^{10}}$ For overviews, see Reinganum (1989) and Gilbert (2006) and for specific models, see Rosen (1991) and Cabral (2003).

¹¹A recent literature focuses on the quality of entrepreneurs, and on their allocation between large and small firms. Hvide (2009) shows that entrepreneurs are more productive, on average, if they work in larger firms, whereas Friebel and Giannetti (2009) work out the implications for talented workers when large firms are more risk-averse, but also reject more good ideas.

quoted in the introduction, we focus on entrepreneurs as providers of major inventions and assume that the incumbent firms do not innovate.¹² The sequence of events in our benchmark model is shown in Figure 1.

******** Figure 1 about here ********

In Stage 1, the entrepreneur makes a fixed monetary investment I in risky R&D, in order to develop an invention. We suppose that there is an infinite number of independent research projects that the entrepreneur may undertake, all requiring the same investment costs I. Hence, investment projects do not vary by the size of the investment, but by the riskiness of the chosen project. Each project k is characterized by a certain success probability p_k . Along the technological frontier, entrepreneurs face a choice between projects that have a high success probability p_k but deliver a small reduction in fixed costs in case of success, and projects that are more risky but also have a larger payoff, if successful. Importantly, we assume that the entrepreneur is riskneutral and thus chooses the project which maximizes the expected net payoff from the investment.¹³

Our benchmark model assumes that a successful invention reduces only the fixed costs of production. This assumption greatly simplifies the exposition, as it implies that product market competition between all firms remains symmetric and the product market price does not depend on the chosen project.¹⁴ To give an example, the fixed cost of producing a prototype part for a new airplane or a racing car can be reduced by small, low-risk improvements in existing technologies. A high-risk, high-return alternative is instead to develop a 3D printer which 'prints' the prototype part using titanium powder, causing virtually no waste of this precious material in the process.¹⁵

¹²Similar results are obtained in a setting where incumbents are simultaneously allowed to innovate, and the entrepreneur can only make a profit when her innovation is successful, while the incumbents' innovation fails. In this setting - which is likely to hold in situations where the incumbents' initial advantage is sufficiently strong - the incumbents' ability to innovate does not affect the entrepreneur's choices, on which the focus of the present analysis lies (see Färnstrand Damsgaard et al., 2010).

¹³Hence, we eliminate the well-known effect that taxes may stimulate entrepreneurial risk-taking by making the government a silent partner in the (risky) operation (Domar and Musgrave, 1944). However, this effect is fully effective only when losses are tax-deductible. Since our analysis explicitly focuses on the limitations of loss offset provisions, the Domar-Musgrave effect is of lesser importance.

 $^{^{14}{\}rm Section}$ 5 considers the more general case where the invention reduces variable production costs.

¹⁵See the article "The printed world" in *The Economist*, 10 February 2011.

With projects differing by their degree of innovation, fixed production costs are

$$F(p_k) = \overline{F} - \Gamma(p_k),\tag{1}$$

where $\Gamma'_k(p_k) < 0, p_k \in (0, 1)$. Omitting the project index, the expected payoff $p\Gamma(p)$ is assumed to be strictly concave in p. Figure 2 illustrates the payoffs of different projects in terms of expected fixed costs reductions.

******** Figure 2 about here ********

As Figure 2 shows, there is a unique project with success probability $0 < \hat{p} < 1$ that maximizes the expected payoff of the invention (or minimizes the expected fixed costs with the invention), given from the first-order condition

$$\Gamma(\hat{p}) + \hat{p}\Gamma'(\hat{p}) = 0.$$
⁽²⁾

In the following, we will refer to an R&D project with a risk level of \hat{p} as the 'costefficient' project. It is instructive to compare the project type chosen by the entrepreneur in equilibrium with this cost-efficient project. More formally, we introduce

Definition 1: The cost-efficient project is given by $\hat{p} = \arg \max_{p} p \Gamma(p)$.

In Stage 2, after investment I has been made and R&D project k has been chosen, the entrepreneur can either sell her invention to one of the incumbents or decide to market the invention herself. If the entrepreneur decides to sell her project, the acquiring incumbent will replace his initial technology with the innovative one. In this case, there will thus still be n firms in the market, though one firm (the acquirer of the innovation) may have a superior technology. In the case where the entrepreneur decides to enter the market, there will be (n + 1) firms in the market, once more with one firm (the entrepreneur herself) having a possibly superior technology.

The entrepreneur's decision of whether to enter the market or sell the innovation to one of the incumbent firms is affected by tax considerations. We denote by t the tax rate faced by the entrepreneur when she decides to enter the market, whereas τ gives the tax rate that is applicable on the income she receives when selling the project to an incumbent firm. We assume that entrepreneurs producing for market entry will incorporate their business and are thus subject to the corporate income tax.¹⁶ In contrast, entrepreneurs producing for sale will usually remain unincorporated and are

¹⁶There is substantial evidence that the larger and the more successful among the new firms are

thus subject to personal income taxation. We assume that there is a tax advantage of market entry so that $t < \tau$.

To motivate this critical assumption in our analysis, Table 1 compares the tax treatment of the income received from selling a business or a patent to the taxation of corporate profits under market entry. Importantly, the sale of a business is classified as a capital gain by some countries, but as ordinary personal business income by others. In most - but not all - of the countries classifying the sale of a business as a capital gain, there will also be a tax advantage for this mode of commercialization. This is true, for example, in the United Kingdom and in the United States.¹⁷ Most countries in continental Europe, however, tax the sale of a business as ordinary income and at tax rates that are substantially above the corporation tax.¹⁸ Entrepreneurs residing in these countries, among them Germany, Italy, and the Nordic countries, thus face a tax advantage when they enter the market themselves, rather than sell their business to an incumbent firm.

A further argument leading to a tax advantage of market entry is that the income tax due under project sale incorporates the value of the entire firm or patent, whereas the profit tax paid under market entry is based only on current-period profits. Hence market entry also acts as a way to postpone the taxation of the valuable asset, and hence to reduce the present value of aggregate tax payments.¹⁹ Finally, many governments subsidize entrepreneurial market entry directly through various support programs at the marketing stage.²⁰

more likely to incorporate (see e.g. Da Rin et al., 2011). In Sweden, for example, only 25% of all firms which started up in 2005 and were still active in 2008 were incorporated (of a total of almost 30,000 start-ups). Among the incorporated start-ups, however, about 72% were high-growth firms, as compared to 34% high-growth firms in other groups (see Tillväxtanalys, 2010).

¹⁷An exception in this category is Ireland where a business sale is classified as a capital gain, but there is still a tax disadvantage relative to market entry, due to the very low Irish corporation tax.

¹⁸This is at least partly due to the fact that corporate tax rates have fallen particularly strongly in Europe over the last decades, as a result of international tax competition. For empirical evidence see, e.g., Devereux et al. (2008).

¹⁹See Landsman and Shackelford (1995) and Ayers et al. (2003) for empirical evidence of this lock-in effect of capital gains taxation.

²⁰In the United States, for example, one of the main programmes to promote small businesses is the Technological Innovation Programme (TIP), which subsidizes the commercialization of successful prototypes with up to USD 3 million. This support scheme is available only if the SME markets the product itself, or is the leading company in a joint venture (OECD, 2010, p. 106). Similar programmes

In Stage 3, the uncertainty is revealed and it turns out whether the innovation is successful or not. If the entrepreneur has not sold her invention, she is free to enter the market at this stage. However, due to entry costs and fixed costs of production, entering the market will only be profitable when the innovation is successful (i.e. fixed production costs are low). If the project fails, the entrepreneur will not enter the market and she will lose all investment costs.

In Stage 4, oligopolistic product market competition occurs between either n or (n+1) firms, depending on the commercialization decision of the entrepreneur in Stage 2 and (in case of market entry) on the success of the project in Stage 3. Equilibrium profits are paid out and taxes are collected on all income.

3 Equilibrium project choice and mode of commercialization

3.1 Stage 4: Product market interaction

We solve the model by backward induction and start with the interaction of firms in the product market. The industry we consider is characterized by fixed costs of production \overline{F} (in the absence of an innovation) and by entry costs G. Free entry to the industry ensures that product market profits (sales revenues minus variable costs) will, in equilibrium, equal the sum of fixed production costs and entry costs. We assume that, in the absence of an innovation that reduces fixed costs, the industry will support n symmetric, incumbent firms. In addition, there is a single entrepreneur, who may enter the market on account of an innovation that reduces fixed production costs.

More specifically, let the set of firms in the industry be $\mathcal{J} = \{i_1, i_2...i_n, e\}$ where $i_1...i_n$ denotes the identical incumbent firms and e is the entrepreneur. The owner of the invention is denoted by $m \in \mathcal{J}$. In the product market interaction, firm j chooses an action x_j , which represents the setting of either a quantity or a price, to maximize its product market profit $\pi_j(x_j, \mathbf{x}_{-j}, m)$. This depends on its own and its rivals' market actions, x_j and \mathbf{x}_{-j} , and on the identity of the owner of the invention, m.

We assume that a unique Nash equilibrium $\{x_j^*(m), x_{-j}^*(m)\}$ exists at this stage, which

exist in many other countries; see OECD (2010) for a summary.

results in positive product market profits. We can then set up a reduced-form for the product market profits of each firm j, taking as given ownership m:

$$\pi_j(m) \equiv \pi_j[x_j^*(m), x_{-j}^*(m), m] \quad \forall \ j \in \mathcal{J}.$$
(3)

Since incumbents are symmetric before the acquisition takes place, we need only distinguish between two types of ownership of the invention: *entrepreneurial* ownership (m = e) and *incumbent* ownership (m = i). Moreover, since the innovation affects only fixed production costs, the product market profit before deduction of fixed costs and taxes is always the same for all active firms in our benchmark model. Hence, there are only two possible levels of such profits: $\pi(i)$ is the profit of each incumbent when the entrepreneur does not enter the market, whereas $\pi(e)$ is the product market profit of incumbents *and* the entrepreneur in case of entry.

We assume that market entry by the entrepreneur will reduce the profit of each producer due to stronger competition, i.e. $\pi(i) > \pi(e)$. This assumption is met in standard models of imperfect competition, such as the oligopoly model of quantity competition in a homogeneous good, or the model of price competition with differentiated products.

As an example, which will be used in our welfare analysis in Section 4, consider quantity competition in differentiated products (Singh and Vives, 1984; Häckner, 2000). The utility of a consumer is then given by:

$$U(\mathbf{q}, Z) = \sum_{j=1}^{n(m)} aq_i - \frac{1}{2} \left(\sum_{j=1}^{n(m)} q_j^2 + 2\gamma \sum_{k \neq j} q_j q_k \right) + Z$$
(4)

where **q** is the vector of outputs, q_i denotes the output of a firm j, Z is a composite numeraire good and a is a constant. The parameter γ measures the substitutability between products, which is inversely related to the level of product differentiation. If $\gamma = 0$, each firm has monopolistic power, whereas if $\gamma = 1$, the products are perfect substitutes. Note finally that the number of firms is n for m = i and (n+1) for m = e. Utility maximization yields inverse demand functions

$$P_j = a - q_j - \gamma q_{-j},\tag{5}$$

where P_j is the price of good j and $q_{-j} = \sum_{k \neq j} q_k$. Quantity competition between firms with constant marginal costs c leads to reduced-form profit expressions for each firm that are quadratic in equilibrium output and given by

$$\pi(m) = [q^*(m)]^2 = \left(\frac{a-c}{2+\gamma[n(m)-1]}\right)^2,\tag{6}$$

thus satisfying $\pi(i) > \pi(e)$.

3.2 Stage 3: Uncertainty revealed

At this stage, it is revealed whether the innovation turns out to be successful or not, where 'success' can either be interpreted in a technological or in a commercial sense. For example, this stage may describe the results of mechanical or medical tests, which determine whether a new, cost-saving technology is feasible. For other innovations, it may be revealed at this stage whether a small-scale market test shows a sufficient acceptance among prospective buyers to make the introduction of the new technology commercially viable.

If the innovation is successful, the superiority of the new product over the existing ones is reflected in reduced fixed costs of $F(p) < \bar{F}$ from (1). Under failure, the invention does not reduce the fixed costs for the owner and fixed production costs remain at \bar{F} . If the owner of the invention is an incumbent firm at this stage, then the success or failure of the innovation has no consequences other than affecting the profits of the acquiring firm. In contrast, if the entrepreneur decided in the previous stage not to sell the invention, the success or failure of the project will affect her decision to enter the market at this stage. Given the free-entry industry equilibrium with n incumbent firms, market entry will be unprofitable for the entrepreneur in case of project failure. To simplify the algebra in the following, it proves convenient to assume that the profits from market entry without an improved technology are just equal to zero for the entrepreneur, but market entry occurs only when profits are strictly positive.²¹ This is formalized in

Assumption A1: When the innovation fails, net profits from entry are zero, $\pi(e) - \overline{F} - G = 0$, and the entrepreneur does not enter. When the innovation is successful, the entrepreneur receives positive net profits equal to $\pi(e) - [\overline{F} - \Gamma(p)] - G = \Gamma(p) > 0$ and enters the market.

In the case of project failure, the entrepreneur's initial investment costs I are lost entirely. To protect the income tax base and prevent fraud, existing tax codes allow

²¹This condition will be approximated when the number of incumbents is not too low. It implies that there is no 'entry hurdle' that would reduce the value of the innovation $(p\Gamma)$ for the entrepreneur in case of market entry.

the deductibility of expenses only in combination with positive income, but do not pay out negative taxes to the taxable entity in case of a loss.²² Moreover, in the case of project failure it is also not possible for the entrepreneur to sell her unused tax credit to one of the incumbents. The reason is that in this case the tax authorities will not accept a link between an incumbent's positive income from existing assets and the losses incurred by the R&D project.

These tax rules imply that there is a tax advantage for the entrepreneur of selling the invention *before* the uncertainty is revealed. This ensures that the entrepreneur can deduct her investment outlays with probability one, because she will then have positive income from the sale with certainty. In contrast, if the entrepreneur were to sell the project only in Stage 3, after the resolution of uncertainty, then a positive sales price would be obtained only when the project is successful. Hence a tax deduction for the investment outlays could also be claimed only with probability p. This is the reason why, in our benchmark model, we assume that a possible project sale occurs before the resolution of uncertainty.²³

3.3 Stage 2: Commercialization

In Stage 2, there is an entry-acquisition game where the entrepreneur decides whether to sell the invention to one of the incumbents or enter the market at the entry cost G, knowing that this is profitable only with a successful project. We assume that the entrepreneur and the incumbents are symmetrically informed about the success probability of the project, and will both base their valuations on the same level of p.²⁴

²²Our static model abstracts from the possibility that the entrepreneur can carry forward the loss for a certain number of years. Empirical evidence suggests that failed start-ups are rarely able to use loss carry-forward provisions in subsequent years. See Auerbach and Altshuler (1990) and Auerbach (2007) for empirical evidence documenting the importance of unused tax credits among U.S. firms.

 $^{^{23}}$ In Section 6 we sketch a model extension where the project can be sold either before or after the uncertainty is lifted. The effects of tax policy in this extended setting are very similar to those in our benchmark case.

 $^{^{24}}$ In the set-up of our model, information asymmetries could work in either direction. While the entrepreneur knows the technical properties of her invention better than incumbents do, incumbents may have more experience in evaluating the likelihood that the innovation succeeds in the market. Given this ambiguity, assuming that the information about p is symmetric represents an intermediate case that has the obvious benefit of analytical simplicity. For an analysis of a project sale where the entrepreneur has an informational advantage, see Norbäck et al. (2010).

The commercialization process is depicted as an auction where n incumbents simultaneously post bids and the entrepreneur either accepts or rejects these bids. If the entrepreneur rejects all bids, she will try to enter the market herself. Each incumbent announces a bid b_i for the invention and $\mathbf{b} = (b_1, ..., b_n)$ is the vector of these bids. Following the announcement of \mathbf{b} , the invention may be sold to one of the incumbents at the bid price, or remain in the ownership of entrepreneur. If more than one bid is accepted, the bidder with the highest bid obtains the invention. If there is more than one incumbent with such a bid, each such incumbent obtains the invention with equal probability.²⁵ The acquisition game is solved for Nash equilibria in undominated pure strategies.

As discussed in Section 2, the entrepreneur faces the tax rate t in case of market entry. We assume that this tax is levied at a proportional rate. Investment costs can be deducted from the tax base when there is positive income, but tax credits are not paid out when the project fails and the entrepreneur's income is thus negative. If the entrepreneur innovates for sale, she will be taxed at the tax rate τ on her capital gains, which are defined as the excess of the sales price over the investment costs. In this case the acquiring incumbent can always deduct the sales price from its positive operating profit, irrespective of whether the invention is successful or not. Using Assumption A1 and defining $\Delta_e(S)$ as the entrepreneur's net gain from selling the invention at price Sover the alternative of market entry gives

$$\Delta_e(S) = \underbrace{S - \tau \left(S - I\right)}_{\text{Net profit from sale}} - \left[\underbrace{p\Gamma(p) - pt[\Gamma(p) - I]}_{\text{Net expected profit from entry}}\right].$$
(7)

From (7), let the reservation price of the entrepreneur be $v_e = \min S$, s.t $\Delta_e(S) \ge 0$. That is, v_e is the minimum price S at which the entrepreneur is willing to sell. Solving for $\Delta_e(S) = 0$, we get:

$$v_e(p) = \frac{(1-t)}{(1-\tau)} \left[p\Gamma(p) - \frac{(\tau - pt)}{(1-t)} I \right] .$$
(8)

²⁵We thus exclude multi-firm licensing in our model. There are several reason why multi-firm licensing might not be an optimal strategy, even if the invention reduces only fixed costs. A first reason is that the sale occurs before the resolution of uncertainty so that no 'product' can be licensed. Secondly, if the licensee and the licensor need to undertake post-licensing investments, free riding problems will likely be increased if there are many licensees. Third, if asymmetric information is present, then multifirm licensing increases the risk that potential buyers can claim that an idea was already known, thus expropriating innovators once they have disclosed their invention to them (Anton and Yao, 1994). These are only some of the reasons why multi-firm licensing rarely occurs in practice.

The reservation price v_e in (8) gives the entrepreneur's product market profits, net of the corporate taxes t that she must pay under market entry, but grossed up by the personal income tax τ that is due under sale.

Next, we turn to the incumbent firms' valuations of the invention. When an incumbent acquires the invention, it is certain that there will only be n firms in the market in the final stage and hence its product market profit is always given by $\pi(i)$. When not acquiring the entrepreneurial firm, the invention can either remain in the hands of the entrepreneur (m = e) or it can be acquired by a rival incumbent firm (m = i). This difference will affect the profits of the non-acquiring incumbent if the invention is successful, because only in this case will the entrepreneur decide to enter the market. When the invention fails, the profit of each incumbent will always be $\pi(i)$ in the product market stage, irrespective of the ownership of the invention. The profits of incumbent firms are taxed at the rate t^{i} .²⁶ Finally, as discussed above, the sales price S is always tax-deductible for the acquiring firm.²⁷ Denoting the net gain for an incumbent firm of acquiring the entrepreneur's invention at a price S by $\Delta_{im}(S)$ for m = e, i yields

$$\Delta_{im}(S) = p(1-t^{i}) \left[\underbrace{\pi(i) - (\overline{F} - \Gamma(p)) - S}_{\text{Profit with project}} - \underbrace{\pi(m) - \overline{F}}_{\text{Profit without project}} \right]$$
Net expected value from a successful innovation
$$+ (1-p)(1-t^{i}) \left[\underbrace{\pi(i) - \overline{F} - S}_{\text{Profit with project}} - \underbrace{\pi(m) - \overline{F}}_{\text{Profit without project}} \right]$$

Net expected value from an unsuccessful innovation

$$= (1 - t^{i}) \left\{ -S + p\Gamma(p) + p[\pi(i) - \pi(m)] \right\},$$
(9)

where we have expanded the right-hand side of (9) with $p\pi(i)$ to arrive at the final expression for $\Delta_{im}(S)$.

 $^{^{26}}$ The tax rate t^i will typically exceed the tax rate t faced by the entrepreneur under market entry, because incumbents are not eligible for reduced tax rates or support schemes tied to small businesses. This difference is immaterial for all of our results, however.

 $^{^{27}}$ Our static model assumes that the full sales price S is immediately tax-deductible for the acquiring firm. In actual tax law, the sales price is treated like an investment that can only be deducted *pro-rata* over several years. Hence, in a dynamic model the present value of being able to deduct S would be somewhat lower for the acquiring firm, but this would not qualitatively alter our results.

From (9), we can define an incumbent firm's valuation as $v_{im} \equiv \max S$, s.t $\Delta_{im}(S) \ge 0$. Solving for $\Delta_{im}(S) = 0$ gives $v_{im} = p\Gamma(p) + p[\pi(i) - \pi(m)]$ as the maximum price S at which an incumbent firm is willing to buy the entrepreneur's invention. Incumbent firms thus have two valuations: The first is a *takeover valuation*, which is an incumbent firm's value of acquiring the invention when this would otherwise remain in the hands of the entrepreneur. In this case m = e and

$$v_{ie}(p) = p\Gamma(p) + p[\pi(i) - \pi(e)],$$
(10)

where $p\Gamma(p)$ is the expected fixed costs savings of the invention and $p[\pi(i) - \pi(e)] > 0$ is the expected increase in product market profits when the entrepreneur is prevented from entering the market.

The second valuation is a *competitive valuation*, which is an incumbent's value of acquiring the invention when a rival incumbent would otherwise obtain it. Then m = iand

$$v_{ii}(p) = p\Gamma(p). \tag{11}$$

Since the invention only affects fixed production costs, the preemptive value in this case is simply the expected fixed cost savings of the invention. Comparing (10) and (11), it is obvious that $v_{ie} > v_{ii}$ since $\pi(i) > \pi(e)$. This describes the concentration effect of an acquisition when entry by the entrepreneur is prevented. Finally, note that the incumbent firms' valuations are unaffected by their profit tax rate t^i , because competitive bidding ensures that the equilibrium sales price will equal the expected increase in profits from acquiring the invention.

We can now proceed to solve for the Equilibrium Ownership Structure (EOS). Since incumbents are symmetric and $v_{ie} > v_{ii}$ always holds, there are three different regimes that we need to consider. The following lemma can then be stated:

Lemma 1 The equilibrium ownership of the invention m^* and the acquisition price S^* are described in Table 2.

Proof: See the Appendix.

Table 2 describes the equilibrium mode of commercialization as a function of the R&D project chosen by the entrepreneur in the first stage, characterized by its success probability p. In Regime 1 (R1 for short), the expected profit from entering the market is higher for the entrepreneur than selling the invention to one of the incumbents. In

Regime	Definition	Ownership	Acquisition price
R1:	$v_e(p) > v_{ie}(p) > v_{ii}(p)$	$m^* = e$	—
R2:	$v_{ie}(p) > v_e(p) > v_{ii}(p)$	$m^* = i$	$S^* = v_e(p)$
R3:	$v_{ie}(p) > v_{ii}(p) > v_e(p)$	$m^* = i$	$S^* = v_{ii}(p)$

Table 2: The equilibrium ownership structure and the acquisition price

Regime 2 the entrepreneur will sell her invention, but the sales price is determined by the reservation price v_e of the entrepreneur. This is because if one incumbent firm bids the reservation price, all other incumbents will only be willing to bid the competitive valuation v_{ii} , which is below v_e in this regime. Hence, the equilibrium bid equals v_e .²⁸ Finally, in Regime 3, the invention is also sold in equilibrium, but the sales price equals the competitive valuation v_{ii} . To see this, note that the reservation price v_e can no longer be the equilibrium price, since the incumbents' competitive valuation v_{ii}

3.4 Stage 1: Project choice by the entrepreneur

In this section, we solve for the equilibrium project selected by the entrepreneur, given that she anticipates the mode of commercialization in the second stage of the game. Noting that investment costs I are independent of project choice, the entrepreneur chooses the project that maximizes the net reward in each regime. From Lemma 1 the net reward for the entrepreneur, denoted by $\Omega(p)$, can be written as

$$\Omega(p) = \begin{cases}
(1-t)p\Gamma - (1-pt)I \equiv (1-\tau)[v_e(p) - I] & \text{in R1} \\
(1-\tau)[v_e(p) - I] & \text{in R2} \\
(1-\tau)[v_{ii}(p) - I] & \text{in R3.}
\end{cases}$$
(12)

Note that in Regime 1 the net reward is the entrepreneur's net expected profit from entry [see eq. (7)] less the investment costs I. By the construction of the reservation

²⁸Note that the gains from an entry deterring acquisition in Regime 2 are unevenly distributed among incumbents, as the acquiring incumbent bears the cost of the entry deterrence while the other firms can free-ride on the acquisition. This raises the possibility of coordination failures among incumbents, if $v_{ie}(p) > v_e(p) > v_{ii}(p)$. We assume, however, that incumbents are able to coordinate by randomly selecting one firm to buy the innovation. Since $v_{ie}(p) > v_e(p)$, this selected firm will then find it profitable to purchase the innovation to prevent entrepreneurial market entry.

price v_e in (8), this is equal to the net reward in the (hypothetical) situation where the entrepreneur receives a sales price v_e and pays personal income taxes on the excess of this sales price over the investment costs *I*. This is also how the net reward is calculated in Regimes 2 and 3, where the sale actually takes place.

To derive the equilibrium project choices we start with Regime 3, where the entrepreneur sells her invention at price $S^* = v_{ii}(p)$. In this regime, the net reward is maximized by incorporating the corporate tax treatment of the *incumbent firms*. From eq. (12), the entrepreneur will choose the project $p_S^* = \arg \max_p (1-\tau)(v_{ii}(p)-I) = \arg \max_p v_{ii}(p)$, where the subscript S stands for the project choice in the sales Regime 3. The associated first-order condition is:

$$\Gamma(p_S^*) + p_S^* \Gamma'(p_S^*) = 0.$$
(13)

Since incumbents can fully deduct the investment costs from their taxable profits, the corporation tax is a lump-sum instrument in this regime. From the perspective of the entrepreneur, the sales price is therefore maximized by choosing the project that maximizes the expected fixed cost reduction, as given by \hat{p} in equation (2).

Next, we consider the optimal project choice in Regimes 1 and 2. In Regime 1 the entrepreneur enters the market herself, whereas in Regime 2 she sells the invention, but the sales price is determined by her reservation price v_e (the expected profits in case of entry). In both regimes, the net reward $\Omega(p) = (1 - \tau)[v_e(p) - I]$ is thus maximized by incorporating the loss offset provisions that apply to the *entrepreneur*. The optimal project is given from $p_E^* = \arg \max_p (1 - \tau)[v_e(p) - I] = \arg \max_p v_e(p)$, where the project choice is denoted by the subscript E. The associated first-order condition is:

$$\Gamma(p_E^*) + p_E^* \Gamma'(p_E^*) = -\frac{t}{(1-t)} I .$$
(14)

The negative term on the right-hand side of (14) shows that in Regimes 1 and 2, the entrepreneur will not choose the cost-efficient project \hat{p} defined in (2). The distortion arises because the entrepreneur cannot deduct her investment costs from tax in the case of project failure. This will induce her to choose a project with an inefficiently low level of risk. This effect is the stronger, the higher is the tax rate t faced by the entrepreneur in case of market entry.

4 The effects of tax policy

4.1 Effects on project choice and commercialization mode

In this section, we analyze how the system of taxing and subsidizing entrepreneurial incomes affects the mode of commercialization and the entrepreneur's project choice. We focus on exogenous variations in the rate of corporate profit taxation that the entrepreneur faces in case of market entry. In this analysis, we hold constant the personal income tax τ , which is levied in case of project sale. To ensure that all possible regimes derived in the preceding section can occur, we assume that market entry must be the entrepreneur's preferred mode of commercialization when t = 0. Effectively, this requires that τ must not be too low. This is formally stated in

Assumption A2: When t = 0, the entrepreneur's reservation value exceeds the incumbents' takeover valuation for any level of p, i.e. $v_e(p)|_{t=0} > v_{ie}(p)|_{t=0}$.

On the other hand, if the entrepreneur faced the same tax rate under the two modes of commercialization, she would always choose to sell her invention to an incumbent firm. This is seen from setting $\tau = t$ in (8), yielding

$$v_e(p)|_{t=\tau} = p\Gamma - \frac{t(1-p)}{(1-t)}I < v_{ii} = p\Gamma, \ \forall p \in [0,1].$$
(15)

In this case, the reservation value of market entry for the entrepreneur falls short of the competitive valuation by incumbents. This implies that, in equilibrium, the entrepreneur sells her invention at the price v_{ii} . By selling the invention, the entrepreneur saves the additional expected tax payments that result from the inability to deduct the investment costs in case of project failure. Since selling the invention yields at least the expected payoff of the invention in the competitive bidding auction modeled here, there are no offsetting benefits from market entry when $t = \tau$ holds.

In the following, we therefore consider corporate tax rates t for the entrepreneur, which range from zero to the personal income tax rate τ . To proceed, we introduce two critical corporate tax rates t^{ED} and t^{C} , where the valuation of the project by the entrepreneur equals the entry deterring (or takeover) valuation and the competitive valuation by the incumbents, respectively. **Definition 2:** Let t^{ED} be defined from $v_e(p_E^*)|_{t=t^{ED}} = v_{ie}(p_E^*)|_{t=t^{ED}}$, and let t^C be defined from $v_e(p_E^*)|_{t=t^C} = v_{ii}(p_S^*)|_{t=t^C}$.

The following proposition describes how the commercialization mode depends on the profit tax rate t.

Proposition 1 (Commercialization mode). In equilibrium, the following holds:

- (i) commercialization by entry (Regime 1) occurs, if the profit tax rate is low, $t \in [0, t^{ED})$;
- (ii) commercialization by sale occurs at the sales price $S^* = v_e(p_E^*)$ (Regime 2), if the profit tax rate is in the intermediate range $t \in [t^{ED}, t^C)$;
- (iii) commercialization by sale occurs at the sales price $S^* = v_{ii}(p_S^*)$ (Regime 3), if the profit tax rate is sufficiently high, $t \in [t^C, 1]$.

The next proposition describes how project choice depends on the profit tax rate.

Proposition 2 (Project choice). In equilibrium, the following holds:

- (i) for low profit tax rates $t < t^C$, the entrepreneur chooses a project with a higher success probability than the cost-efficient one, $p^{opt} = p_E^* > \hat{p}$;
- (ii) for sufficiently high profit tax rates $t \ge t^C$, the entrepreneur's project choice is efficient, $p^{opt} = p_S^* = \hat{p}$.

To prove Propositions 1 and 2, it is convenient to first evaluate the net reward $\Omega(p)$ in equation (12) at the optimal project p^{opt} . Inserting the results from (13) and (14) gives:

$$\Omega(p^{opt}) = \begin{cases} (1-\tau)[v_e(p_E^*) - I] & \text{in R1} \\ (1-\tau)[v_e(p_E^*) - I] & \text{in R2} \\ (1-\tau)[v_{ii}(p_S^*) - I] & \text{in R3.} \end{cases}$$
(16)

The entrepreneur will choose the R&D project and the mode of commercialization to maximize the net reward $\Omega(p^{opt})$. From (16), we note that this is equivalent to maximizing the reward from the commercialization auction in Lemma 1. Thus, the entrepreneur will choose the optimal project p_E^* under market entry (or under sale at the reservation price), if the reward to this project exceeds the reward from choosing the optimal project p_S^* which induces a sale under bidding competition, $v_e(p_E^*) > v_{ii}(p_S^*)$.

****** Figure 3 about here *****

Panel (i) of Figure 3 depicts the reservation price $v_e(p_E^*)$ and the competitive valuation of incumbents $v_{ii}(p_S^*)$ as functions of the tax rate t. Note that the reservation price $v_e(p_E^*)$ is decreasing in t, as is seen from totally differentiating (8). At the same time, the incumbents' competitive valuation, evaluated at the optimal project choice p_S^* , is unaffected by taxes from (11). Hence there must exist a critical tax rate t^C such that $v_e(p_E^*)|_{t=t^C} = v_{ii}(p_S^*)|_{t=t^C}$. As shown in panel (i), when $t > t^C$ the entrepreneur will always choose p_S^* to induce a subsequent sale under bidding competition. If $t < t^C$ instead, she will choose a project p_E^* that maximizes the reservation price obtained either from commercialization by entry or from an entry-deterring acquisition.

We can now prove Proposition 2. From the above argument, the equilibrium project p^{opt} is given by

$$p^{opt} = \begin{cases} p_E^*, \ t \in [0, t^C), \\ p_S^*, \ t \in [t^C, 1]. \end{cases}$$
(17)

Implicitly differentiating (13) in Regime 3 and (14) in Regimes 1 and 2 then yields

$$\frac{dp^{opt}}{dt} = \begin{cases} \frac{-I}{(2\Gamma' + p_E^*\Gamma'')(1-t)^2} > 0 & t \in [0, t^C) \text{ in R1, R2,} \\ 0 & t \in [t^C, 1] \text{ in R3,} \end{cases}$$
(18)

where $2\Gamma' + p_E^* \Gamma'' < 0$ follows from the assumption that $p\Gamma(p)$ is strictly concave.

Panel (ii) of Figure 3 illustrates the relationship between the corporate tax rate t and the equilibrium project choice, p^{opt} , as given in (18). As long as $t < t^C$, raising the profit tax rate leads to a continuous increase in the success probability of the equilibrium project, because the rise in t makes the deductibility of the initial investment outlays more valuable, and this deductibility can only be used when the project is successful. At $t = t^C$, the entrepreneur sells the invention at price $S^* = v_{ii}$. At this point the success probability of the equilibrium project drops back to the efficient level $p_S^* = \hat{p}$, and it remains independent of t as the profit tax rate rises further. Overall, therefore, the equilibrium level of p adjusts in a non-monotonous way to the profit tax rate t.

Turning to the proof of Proposition 1, we can now apply Lemma 1 to the optimal projects p^{opt} given from Proposition 2. In panel (iii) of Figure 3, we depict the locus of the valuations v_e , v_{ie} and v_{ii} as reduced-form functions of the profit tax rate t, incorporating the equilibrium choices of p^{opt} that apply at each level of t. To determine

the shape of the different curves, we totally differentiate (8), (10) and (11). This yields:

$$\frac{dv_r(p^{opt})}{dt} = \underbrace{\frac{\partial v_r}{\partial t}}_{\text{direct effect}} + \underbrace{\frac{\partial v_r}{\partial p} \frac{dp^{opt}}{dt}}_{\text{indirect effect}} \quad \forall r \in \{e, ie, ii\}.$$
(19)

We turn first to the entrepreneur's reservation price v_e . By the direct effect, this is falling in the profit tax rate t across all regimes. The indirect effect operating through changes in the optimal project choice is zero for $t < t^C$ due to the envelope theorem. At $t = t^C$ the equilibrium project choice changes discretely to p_S^* . This leads to a discontinuous downward jump in v_e , as the equilibrium project no longer maximizes the entrepreneur's reservation price.²⁹

An analogous argument applies for the valuations of incumbents. The direct effect of t on both v_{ie} and v_{ii} is zero from equations (10) and (11). The indirect effects in eq. (19) consist of the induced changes in project choice, as given in (18), and the incumbents' valuation of these changes, $\partial v_{im}/\partial p$. The latter effects are zero in Regime 3 from the envelope theorem. In Regimes 1 and 2, these effects are given by

$$\frac{\partial v_{ie}}{\partial p} = \Gamma + p_E^* \Gamma' + \pi(i) - \pi(e), \qquad \frac{\partial v_{ii}}{\partial p} = \Gamma + p_E^* \Gamma'.$$
(20)

From equation (14), we know that the second of these terms is negative. Hence increases in the tax rate t (and thus in p_E^*) have a negative effect on the incumbents' competitive valuation v_{ii} for $t < t^C$. In contrast, the effect on the takeover valuation v_{ie} is ambiguous, due to the additional term $\pi(i) - \pi(e) > 0$. At $t = t^C$, there is an upward jump in the graphs of both v_{ii} and v_{ie} , because the discrete switch in the equilibrium project choice then maximizes the valuations of incumbents. Finally, note that since there exists a critical tax level t^C where $v_e(p_E^*)|_{t=t^C} = v_{ii}(p_S^*)|_{t=t^C}$, there must also exist a second critical level t^{ED} where $v_e(p_E^*)|_{t=t^{ED}} = v_{ie}(p_E^*)|_{t=t^{ED}}$ holds. Moreover, since $v_{ie}(p^{opt}) - v_{ii}(p^{opt}) = \pi(i) - \pi(e) > 0$, it also follows that $t^{ED} < t^C$.

We are now ready to prove Proposition 1. When the corporate tax rate is low, $t \in (0, t^{ED})$, the entry value $v_e(p_E^*)$ exceeds the incumbents' takeover valuation $v_{ie}(p_E^*)$ from Assumption A2. This leads to an equilibrium in Regime 1 with the entrepreneur retaining the ownership of her invention and entering the market in case the invention succeeds. As the profit tax rate increases, it reaches the first critical value t^{ED} , where

²⁹Technically, the envelope theorem can no longer be applied at $t = t^C$, because the equilibrium project is then not chosen to maximize v_e .

the entrepreneur's reservation value equals the takeover valuation of the incumbents. At $t = t^{ED}$, the equilibrium switches to Regime 2 with an entry deterring acquisition taking place at the acquisition price $S^* = v_e(p_E^*)$. As t rises further, it reaches the second critical level t^C , where the entrepreneur's reservation value falls below the competitive valuation of the incumbent firms, $v_e(p_S^*) < v_{ii}(p_S^*) < 0$. This induces a bidding war between incumbents and results in Regime 3, where the sales price is fixed by the incumbents' valuation $v_{ii}(p_S^*)$. Further increases in t continue to reduce the entry value of the entrepreneur, which falls to its minimum value at $t = \tau$. The equilibrium commercialization pattern is summarized at the bottom of panel (iii). \Box

To summarize, government policies that reduce the profit taxation of small businesses encourage market entry by entrepreneurs and foster competition in the innovative sector. At the same time, however, the entrepreneur's choice of project will be distorted whenever she produces for market entry. As is shown in panel (iii) of Figure 3, when profit tax rates are in the range $t \in [0, t^C)$, even risk-neutral entrepreneurs will choose projects that involve too little risk and fall short of maximizing the expected return from the investment. This effect arises from the imperfect loss offset that entrepreneurs face in case of project failure. We now turn to the welfare implications of this result.

4.2 Welfare and employment effects of tax policy

In this section we analyze how taxes impact expected welfare and employment by affecting the R&D project choice and the commercialization mode of the entrepreneur. To this end we measure welfare as the sum of aggregate industry profits, consumer surplus and tax revenue. In Regime 1, where the entrepreneur enters the market, the after-tax net reward is given by $(1 - t)[p_E\Gamma(p_E)] - (1 - pt)I$ in eq. (12). The expected income loss for the entrepreneur that arises from the inability to deduct investment cost from tax in case the project fails is, however, fully compensated by higher expected tax payments, $tp(\Gamma - I)$. Hence the sum of tax revenue and the entrepreneur's net reward equals the gross-of-tax reward from the innovation, net of investment costs, $p_E\Gamma(p_E) - I$. In Regimes 2 and 3, where the innovation is sold to an incumbent, the personal income tax τ is applied on the tax base $(v_e - I)$ and $(v_{ii} - I)$, respectively. Together with tax revenues, the social surplus is thus again $p_E\Gamma(p_E) - I$ in Regime 2 and $p_S \Gamma(p_S) - I$ in Regime 3. This yields the following expressions for expected welfare:

$$E[W] = \begin{cases} p_E^* \Gamma(p_E^*) - I + n\pi(i) + CS(i) + p_E^* \xi & \text{in R1} \\ p_E^* \Gamma(p_E^*) - I + n\pi(i) + CS(i) & \text{in R2} \\ p_S^* \Gamma(p_S^*) - I + n\pi(i) + CS(i) & \text{in R3.} \end{cases}$$
(21)

Here

$$\xi \equiv n \left[\pi(e) - \pi(i) \right] + CS(e) - CS(i) \leq 0$$

is the net externality arising from the market entry of the entrepreneur in Regime 1, which increases the equilibrium number of firms from n to (n + 1). This causes a decline in the incumbents' aggregate profits given by $n[\pi(e) - \pi(i)] < 0$ and an increase in consumer surplus given by CS(e) - CS(i) > 0. Employing the model of quantity competition in differentiated products introduced in Section 3.1 [see eq. (4)] yields

$$CS(m) = aQ^{*}(m) - \frac{1}{2} \left[n(m)(q^{*}(m))^{2} + \gamma n(m)[n(m) - 1](q^{*}(m))^{2} \right] -n(m) \left[a - q^{*}(m) - \gamma (n - 1)q^{*}(m) \right] q^{*}(m) \quad \forall m \in \{i, e\},$$
(22)

where $Q = \sum_{j} q_{j}$ and n(m)[n(m) - 1]/2 is the number of unique cross-products terms $q_{j}^{*}q_{k}^{*}$ with $j \neq k$ in (4). From (22) and (6), the net externality ξ becomes

$$\xi = \frac{1}{2}(a-c)^2 \frac{\left[-n^2\gamma^3 - n^2\gamma^2 + n\gamma^3 - 3n\gamma^2 + \gamma^2 - 4\gamma + 4\right]}{\left(n\gamma + 2\right)^2 \left(-\gamma + n\gamma + 2\right)^2},$$
(23)

where n is the number of incumbent firms. From (23) it is straightforward to infer that $\xi > 0$ for $\gamma = 0$, but $\xi < 0$ for $\gamma = 1$. Moreover, since ξ is continuous in γ , there must exist a critical level of product differentiation $0 < \gamma^c < 1$ such that $\xi = 0$.

Hence, if the goods produced by the incumbents and the entrepreneur are perfect substitutes ($\gamma = 1$), then entrepreneurial market entry has a negative net effect on surplus in the industry, because the reduction in producer surplus dominates the rise in consumer surplus. This is due to the additional fixed costs incurred by the new entrant, which leads to rising average costs in the industry (Mankiw and Whinston, 1986). If all firms are local monopolies, however ($\gamma = 1$) then the increase in consumer surplus will dominate, as consumers value the addition of a new product.

We can now discuss the overall welfare effects of tax policy. If $\xi < 0$ holds in Regime 1, then market entry by the entrepreneur unambiguously reduces aggregate welfare, relative to the alternative of selling the innovation. Recall that $p_E\Gamma(p_E)$ in Regimes 1 and 2 is below $p_S\Gamma(p_S)$ in Regime 3, because the entrepreneur's project choice is distorted in the first two regimes, but not in the third [eqs. (13) and (14)]. In this case, welfare is unambiguously highest in Regime 3, implying that the optimal policy is to tax the entrepreneur's profits under market entry at the same rate as the proceeds from the sale of the innovation to an incumbent firm.

In contrast, when $\xi > 0$, then an argument arises to tax-discriminate in favour of entrepreneurial market entry, as this will benefit consumers due to the additional variety. At the same time, market entry by the entrepreneur will imply a project choice that involves an inefficiently low level of risk-taking. This distortion of project choice will be the more severe, the larger is the (positively defined) elasticity with which the cost savings Γ respond to the success probability of the project, $\eta = -(\partial \Gamma/\partial p)(p/\Gamma)$. Hence the optimal tax policy faces a trade-off between the net benefits from entrepreneurial entry that result in Regime 1, and the benefits from an efficient project choice that arise in Regime 3. A tax policy leading to an equilibrium in Regime 2 is never optimal, because this regime features neither entrepreneurial market entry nor an efficient project choice. We summarize these results in:

Proposition 3 (Welfare effects of tax policy). Consider a model of Cournot competition in differentiated products. Then the following holds:

- (i) if products are sufficiently close substitutes (γ → 1), the optimal tax policy induces an equilibrium in Regime 3, with commercialization by sale at the competitive valuation of incumbents.
- (ii) if products are strongly differentiated (γ → 0), the optimal tax policy induces an equilibrium in Regime 1, with entrepreneurial market entry, when the net externality ξ > 0 is sufficiently large, relative to the elasticity of cost savings η. Otherwise, the optimal tax policy induces an equilibrium in Regime 3.

The argument for promoting entrepreneurial market entry is strengthened when there are market failures in labour markets and countries suffer from unemployment. As we have discussed in the introduction, one of the main objectives of supporting small, entrepreneurial firms is the belief that they create additional jobs. It is straightforward to incorporate this argument into our analysis when we assume that aggregate employment in the industry is monotonously rising in aggregate output. The market entry of the entrepreneur will not displace incumbent firms, because the entry costs G are sunk for them. Hence, entrepreneurial market entry will increase aggregate output in the industry.³⁰ Moreover, given macroeconomic unemployment, a higher employment level in the innovative industry is unlikely to be fully compensated by employment reductions in other parts of the economy.

Summing up, a policy of promoting market entry of small, innovative firms is likely to lead to higher output and employment in the innovative sector, but the welfare effects of this policy are positive only when goods are sufficiently differentiated. The costs of an entry-promoting policy lie in the distortion of the entrepreneur's project choice, in the direction of inefficiently low levels of risk-taking. In our benchmark model, where the innovation affects only fixed production costs, this points to a basic policy trade-off between the declared goals of promoting employment in small, innovative businesses and the fostering of major, risky innovations. The next section analyzes how the effects of an entry-promoting tax policy change when the choice of the R&D project also affects variable production costs.

5 Variable cost saving inventions

In this section, we show that if more risky projects lead to lower variable costs, rather than lower fixed costs, consumers may unambiguously prefer commercialization by sale over commercialization by entry. For this purpose, we briefly discuss how the analysis in the different stages changes when variable cost reductions are allowed for.

Stage 4: Consider a situation where the gains from a more risky project in case of success are given by larger variable cost savings for the possessor of the invention, while fixed costs are ignored. Let the acquiring incumbent's product market profit for a successful invention be $\pi_A(i, p)$, where p is the project choice in Stage 1. Similarly, let the entrepreneur's profit when entering be $\pi_E(e, p)$ and let a non-acquiring incumbent's profit be $\pi_{NA}(m, p)$. Note that we write these profits as reduced-form functions of the project choice p, indicating that the project has succeeded in Stage 3.

We then introduce:

Assumption A3: (i)
$$\frac{d\pi_A(i,p)}{dp} < 0, \frac{d\pi_E(e,p)}{dp} < 0$$
, (ii) $\frac{d\pi_{NA}(m,p)}{dp} > 0, m \in \{e,i\}$.

³⁰This is easily verified for the Cournot model with differentiated outputs by multiplying the perfirm outputs in eq. (6) for m = i, e with n and (n + 1), respectively.

Assumption A.3 (i) states that the product market profit from a successful invention is smaller for the possessor (either the acquiring incumbent or the entrepreneur herself) when the riskiness of the project decreases. Assumption A.3 (ii) states that non-acquiring incumbents see their profits increasing when the possessor has a safer project, since rivals then face less fierce competition from the owner of the invention.³¹

Stage 3: At this stage, it is revealed whether the innovation turns out to be successful. We maintain Assumption A1 so that entry is profitable only if the project succeeds. Hence $\pi_E(e, p) - G \ge 0$, but $\pi_E(e, 0) - G = 0$, where $\pi_E(e, p)$ is the profit of the entrepreneur under entry with project p and $\pi_E(e, 0)$ is the profit with a failed project.

Stage 2: At the commercialization stage, the entrepreneur's reservation price defined in (8) becomes:

$$v_e = \frac{(1-t)}{(1-\tau)} \left\{ p[\pi_E(e,p) - G] - \frac{(\tau - pt)}{(1-t)} I \right\}.$$
 (24)

The takeover and the competitive valuation of an incumbent in (10) and (11) become

$$v_{ie} = p \left[\pi_A(i, p) - \pi_{NA}(e, p) \right], \qquad v_{ii} = p \left[\pi_A(i, p) - \pi_{NA}(i, p) \right], \tag{25}$$

where again $v_{ie} > v_{ii}$ since $\pi_{NA}(e, p) < \pi_{NA}(i, p)$. From the latter inequality, it follows that the equilibrium commercialization mode can be solved by applying Lemma 1.

Stage 1: Turning to the entrepreneur's project choice, we assume that $p\pi_E(e, p)$ and $p[\pi_A(i, p) - \pi_{NA}(m, p)]$ are strictly concave in p, ensuring well-defined project choices. Introducing $E[\pi(m, p)] \equiv p\pi(m, p)$ as the expected value of the project and using Assumption A1, the first-order condition for the optimal project choice when innovating for entry or selling at the reservation price $S^* = v_e$ in Regimes 1 and 2 becomes:

$$\frac{dE[\pi_E(e,p)]}{dp} = 0 \quad \Longrightarrow \quad p_E^* \frac{d\pi_E(e,p_E^*)}{dp} = -\frac{t}{(1-t)} I . \tag{26}$$

As in our benchmark model [see eq. (14)], the inability to deduct the investment costs from tax in case of failure will induce the investor to choose a project with too little risk, other things being equal.

³¹To illustrate, let c be the incumbents' initial marginal cost. If the entrepreneur's project succeeds in Stage 3, the marginal cost falls to $c_E(p) = c - (1 - p)$ in Stage 4, which is also an acquiring incumbent's marginal cost, $c_E(p) = c_A(p)$. Non-acquiring incumbents continue to have marginal costs of $c_{NA} = c$. Assuming linear demand and assuming Nash-Cournot competition, it is routine to show that the reduced-form profits fulfill Assumption A3.

When innovating for sale under bidding competition in Regime 3, receiving the sale price $S^* = v_{ii}$ [see eq. (13)], the optimal project choice is given by

$$\frac{dE[\pi_A(i,p)]}{dp} - \frac{dE[\pi_{NA}(i,p)]}{dp} = \pi_A(i,p_S^*) + p_S^* \frac{d\pi_A(i,p_S^*)}{dp} - \pi_{NA}(i,p_S^*) - p_S^* \frac{d\pi_{NA}(i,p_S^*)}{dp} = 0$$
(27)

The optimal project choice in Regime 3 is again independent of the tax rate. There is, however, an important difference to our benchmark case. With variable cost reductions, an entrepreneur that chooses an optimal project for sale will not only consider how the expected product market profit of the acquirer is affected, but she will also take into account that choosing a safer project increases the expected profit for a non-acquirer (see Assumption A3). Since the incumbents' willingness to pay for the project is negatively affected by the profits of a non-acquirer [see eq. (25)], this gives a strategic increase a more risky project.

This strategic incentive is shown in the lower panel (ii) of Figure 4, where the slope of the marginal expected profit curve from a change in p is always steeper in Regime 3 as compared to Regimes 1 and 2. For this reason the difference in the risk characteristics of the projects chosen in Regimes 1 and 2 on the one hand and in Regime 3 on the other is even larger than in our benchmark case.

******* Figure 4 about here ********

The effects of tax policy on consumers. We now examine how tax rates affect consumers through the entrepreneur's choices of project and mode of commercialization. Maintaining Assumption A2, we proceed as in Section 4 and define reduced-form valuations $v_r(t) \equiv v_r[p^{opt}(t)]$. Taking the total derivative in t and applying the envelope theorem, it is straightforward to show that Propositions 1 and 2 continue to hold when more risky projects are associated with larger variable cost reductions.

Consider now the upper panel (i) of Figure 4. Let \overline{CS} be the consumer surplus when the invention has failed. The expected consumer surplus under innovation by entry and under innovation by sale is then $E[CS(e, p)] \equiv pCS(e, p) + (1-p)\overline{CS}$ and $E[CS(i, p)] \equiv$ $pCS(i, p) + (1-p)\overline{CS}$, respectively. For the same project p, innovation by entry always gives a higher expected consumer surplus since CS(e, p) > CS(i, p) from the concentration effect of an acquisition. Assume that the expected consumer surplus is strictly concave in p, so that there exist optimal projects, from the perspective of consumers, given by $p_S^{CS} = \arg \max_p [pCS(p, i) + (1-p)\bar{C}S]$ and $p_E^{CS} = \arg \max_p [pCS(p, e) + (1-p)\bar{C}S]$. Note that, because of imperfect competition in the product market, the interests of producers and consumers are generally not aligned in our model. Regardless of entry mode, the entrepreneur will therefore not choose a project that maximizes the expected consumer surplus, so that in general $p_E^* \neq p_E^{CS}$ and $p_S^* \neq p_S^{CS}$ holds.

Suppose that we start from a high corporate tax rate for the entrepreneur, $t > t^C$. From Proposition 1, this implies that the entrepreneur will choose commercialization by sale at the sales price $S^* = v_{ii}$. This yields an expected consumer surplus of $E[CS(e, p_S^*)]$, as shown by point S in Figure 4 (i). Suppose then that the tax rate is reduced to $t_1 < t^C$ so that the entrepreneur chooses instead commercialization by entry. With the new tax rate t_1 the entrepreneur will choose an overly safe project under market entry, due to the incomplete loss offset provisions of the corporate tax code. This project choice may yields only limited reductions in variable costs, and hence consumer prices, as given by the point E_1 in Figure 4 (i). A comparison of the points S and E_1 then reveals that the expected consumer surplus can be lower under market entry than under sale, even though the number of competitors is higher with market entry by the entrepreneur.

In this case the interests of consumers will be best served when the entrepreneur sells the innovation to an incumbent firm in Regime 3 and chooses a project that minimizes the expected variable costs of production. Moreover, commercialization by sale will also be preferred from the perspective of creating new jobs, as the lower consumer prices will generally be associated with higher total output and employment. We conclude, therefore, that in a setting where innovations reduce variable production costs, a policy of promoting the market entry of entrepreneurs by means of tax advantages can be counterproductive from *both* an employment perspective and from the viewpoint of fostering highly productive innovations.

6 Discussion

In this section, we briefly discuss several further extensions and modifications of our benchmark model.³²

 $^{^{32}}$ A detailed analysis of the first three extensions is found in the supplementary appendix on our homepages.

Post-uncertainty sales. As we have discussed in Section 3.2, there is a basic tax advantage for the entrepreneur, if she sells before the uncertainty is lifted. This is why our benchmark model has ruled out the possibility of a project sale after the resolution of uncertainty. In an extension we pursue the implications for the model's results, if this assumption is relaxed and both pre- and post-uncertainty sales are permitted. This analysis becomes rather involved when we allow for independent draws of the incumbents selected to buy the innovation at the pre- and post-uncertainty stage (cf. footnote 28). In this case the incumbent drawn at the pre-uncertainty stage has an incentive not to buy the innovation and rely instead on the possibility that he is not drawn as the buyer at the post-uncertainty stage (provided that the innovation has been successful). As a result, post-uncertainty sales may occur in this extended model for some range of tax rates. In effect, this is a dynamic coordination failure as the incumbent who ends up buying the innovation post-uncertainty would have preferred a pre-uncertainty sale, in order to benefit from the tax advantage of the latter.

When is this incentive to free-ride on a late rival's acquisition large enough to generate a situation where incumbents will abstain from an early acquisition? To explore this, we provide here the main intuition for the result and refer the reader to the web appendix for all details. A first result is that an equilibrium with a sale in Stage 3 is incompatible with bidding competition and hence cannot lie in Regime 3. Hence the relevant sales price is always the reservation price of the entrepreneur. We thus evaluate the net value for an incumbent of an early acquisition at the entrepreneur's reservation price $S^{2^*} = v_e(p_E^*)$. Incumbents know that waiting may imply a new acquisition game in Stage 3, where the probability of being drawn as acquirer is 1/n and the late acquisition price is $S^{3^*} = w_e(p_E^*) = \frac{(1-t)}{(1-\tau)} \left[\Gamma(p_E^*) - \frac{(\tau-t)}{(1-t)}I\right]^{33}$

Evaluating the net value of an early acquisition in Stage 2 gives:

$$v_{ie}(p_E^*)|_{S^{3*}=w_e} - v_e(p_E^*) = (1 - p_E^*) \frac{\tau}{(1 - \tau)} I - p_E^* (1 - 1/n) \left[w_e(p_E^*) - w_{ii}(p_E^*) \right].$$
(28)

The incentive for an incumbent to buy early then consists of the two terms on the right-hand side of eq. (28). The first of these terms measures the expected loss for the entrepreneur in terms of foregone tax deductions when the project fails. This effect arises because with a sale in Stage 3 the deduction can only be claimed in case of success, whereas a sale in Stage 2 implies that the tax deduction can be claimed with

³³Here the symbol w stands for project valuations in Stage 3. An incumbent's valuation of a successful innovation in Stage 3 involves no uncertainty and is given by $w_{im}(p) = \pi(i) - \pi(m) + \Gamma(p)$.

certainty. Other things equal, this will lead to a 'price rebate' for an early acquisition, as the entrepreneur will accept a lower price in Stage 2 to ensure the tax deduction for the initial outlay I. This deduction effect thus pushes incumbents towards an early acquisition. The second term on the right-hand side of (28) is a free-riding effect. This effect reduces an incumbent's willingness to pay in Stage 2, since he can gamble on the possibility that a rival incumbent will conduct a late acquisition in Stage 3, thus allowing the incumbent to avoid the 'excess payment' $w_e - w_{ii} > 0$.

The free-riding effect can dominate the deduction effect when the tax rate t is sufficiently high. This follows from the fact that the entrepreneur will choose safer projects when the tax rate increases, $dp_E^*/dt > 0$. Safer project choices then reinforce the freeriding effect while dampening the deduction effect. This may lead to a range of tax rates where the entrepreneur develops projects which she will not sell initially in Stage 2, but rather wait and sell in Stage 3, if the project turns out to be successful.

Importantly, however, even if post-uncertainty sales occur, the equilibrium project choices are unchanged from our benchmark analysis for all possible tax rates and Proposition 2 continues to hold. In particular, low tax rates for entrepreneurial market entry $(t < t^{C})$ will still lead to project choices with inefficiently low levels of risk-taking, whereas sufficiently high tax rates for the market entry of the entrepreneur $(t > t^{C})$ will lead to efficient project choices. Hence, allowing for post-uncertainty sales does not change the fundamental trade-off for tax policy derived in our benchmark model.

Effort by the entrepreneur. An important aspect of entrepreneurial innovation is that a substantial share of the initial investment may consist of effort put in by the entrepreneur. It is straightforward to incorporate this into our model by introducing a zero stage of the game, where the entrepreneur chooses an endogenous level of effort in order to generate a basic innovative idea. In this setting it can be shown that increased levels of taxation reduce the entrepreneur's effort to create innovative ideas. While the effort-reducing effects of taxes have been stressed in previous literature (e.g. Keuschnigg and Nielsen, 2004), this result is specified in our setting because different taxes affect entrepreneurial effort under different equilibrium modes of commercialization. In particular, higher corporate tax rates are effort-reducing when the equilibrium is in Regimes 1 and 2, whereas higher income taxes reduce effort when the equilibrium is in Regime 3. All other results of our analysis are unaffected by this model extension. Incorporation and transaction cost for sale. Another modification is to permit entrepreneurs that innovate for sale to incorporate, in order to benefit from lower corporate tax rates. For analytical simplicity, we focus on the extreme case where tax rates are identical when innovating for market entry or for sale, thus ignoring the fact that some government subsidies for SMEs are only available when entrepreneurs actually enter the market. In this setting the trade-off between market entry and sale arises from the fact that the sale of the innovation is subject to a fixed transaction cost. Market entry by the entrepreneur will thus be induced by the motive to save on transaction costs, rather than by tax advantages. It remains true, however, that the entrepreneur chooses an inefficiently low level of risk whenever she innovates for market entry, as a result of incomplete loss offset provisions.

It could further be argued that the transaction cost of a sale to incumbents is not fixed, but is rising in the degree of project risk. This could be due to potential moral hazard problems that arise from an information advantage that the entrepreneur has over the incumbents with respect to the characteristics of her innovation. Other things being equal, this risk-dependent transaction cost would then lead to a project choice with a lower level of risk-taking. At the same time, however, entry costs can also be argued to rise in the riskiness of the innovation project, due to the entrepreneur's lack of knowledge of market conditions (cf. footnote 24). As a result, the entrepreneur may opt for less risk under both modes of commercialization when these additional model elements are incorporated. In any case it remains true that incomplete loss offset provisions are an additional reason to reduce project risk under market entry, which is not present when the innovation is sold to an incumbent.

Product innovation. Our model is based on a setting where the project chosen by the entrepreneur is a cost-reducing process innovation. It is conceptually straightforward, however, to alternatively consider a product innovation where the entrepreneur develops a new product in the first stage. The degree of innovation, and the risk associated with it, would then both rise in the distance that the new product has to existing ones as, for instance, in Salop's (1979) circular-city model of product differentiation. While this model would be more complex than the one used here, a concave expected return curve similar to the one in Figure 2 could also be obtained in this case. All other features of our model would also be preserved in such a changed setting. In particular, imperfect loss offset would again cause the entrepreneur to choose a project with too

little risk, which would here imply a project that is "too similar" to existing ones.

Venture capital. Finally, we discuss in some more detail the effects of incorporating venture capitalists into our model. Our benchmark analysis has assumed that the entrepreneur is not able to obtain tax rebates for the investment costs incurred when she attempts to enter the market herself, but the project fails. These loss offset restrictions imply that the entrepreneur could gain from the cooperation with a financial partner, who has other income to deduct the losses against. In this case the investment cost I could be deducted as long as the partner has sufficient positive income, and the distortionary effect on project choice could thus be avoided. It is well known, however, that such financial partnerships are associated with severe adverse selection and moral hazard problems. Therefore, where financial partners exist, they almost always involve a highly specialized venture capitalist.

In most countries, however, only few projects are backed by venture capital. Even in the United States, where the venture capital market is by far the largest, only 14,000 portfolio companies worldwide received venture capital over a 30-year period from 1975 to 2005 (Lerner, 2010). Total venture capital investments in the United States amounted to 28.8 billion US-\$ in 2008, corresponding to 0.20 percent of this country's GDP.³⁴ In Europe, the share of venture capital investments in GDP is much lower, equalling 0.07 percent of national GDP in Sweden, 0.05 percent in the United Kingdom and 0.03 percent of GDP in Germany (EFI, 2011, p. 19).

Moreover, even for start-up firms backed by venture capital (VC), the possibility to deduct losses for tax purposes is far from complete. On the one hand, entrepreneurs usually retain a substantial ownership share in the VC-backed company, in order to mitigate the moral hazard problems inherent in the relationship with the venture capitalist (see Keuschnigg and Nielsen, 2002, 2004). This implies that loss offset restrictions will still apply for the ownership share of the entrepreneur. In addition, some countries' tax laws also restrict the tax deduction of losses for venture capitalists. In Germany, for example, unused tax losses cannot be transferred when the venture capitalist sells her shares in the company - for example to another VC firm, which specializes in financing a different development phase of the start-up. These restrictive tax provisions are widely believed to hinder VC financing, but also the establishment and growth of innovative companies in Germany (EFI, 2011, p. 19).

³⁴See "Investment funding rose 5 percent in 2008", VentureBeat, 18 February 2009.

In sum, bringing venture capital financing into the picture and accounting for its relative importance will introduce some qualifications to our assumption that an entrepreneur producing for market entry is not able to obtain any tax rebates for losses incurred in the investment phase. However, this extension will clearly not overturn our basic argument that loss offset restrictions are more severe when the entrepreneur produces for market entry, as opposed to selling her innovation to an incumbent firm.

7 Conclusion

In this paper we have focused on two important decision margins of entrepreneurs that have received little analysis so far in a context of tax policy. These are the entrepreneur's choice between innovations with different risk and return characteristics, and her decision of how to commercialize the innovation. In a setting where innovations affect only fixed production costs, we have shown that tax policies to support small, technology-intensive businesses developments can be welfare-increasing if and only if goods in the innovative sector are sufficiently differentiated. Moreover, these policies face a fundamental trade-off: they promote market entry by entrepreneurs and increase aggregate industry output, but at the same time lead to R&D choices that involve suboptimally low levels of risk-taking. This is due to the risk-reducing effect of limited loss offset provisions, which are particularly important for small, entrepreneurial firms. In that sense, the comparative advantage of new, entrepreneurial enterprises to create major innovations is thus compromised by such a policy. Inefficient project choices will have even more severe implications when innovations reduce marginal production costs. In this case it is possible that policies to induce entrepreneurial market entry are unambiguously harmful, reducing both output and employment, in addition to distorting the choice of the innovation project.

These results can be contrasted with existing policies to support entrepreneurship in the European Union and elsewhere. The Small Business Act for Europe (European Commission, 2008), for example, simultaneously aims at fostering risky innovations and employment growth by promoting the market entry of small, entrepreneurial enterprises. Our analysis shows that these goals are mutually incompatible, at best, when the interaction between the entrepreneur's project choice and the choice of commercialization mode is explicitly analyzed. Instead, the results from our model suggest to reduce or even eliminate the policy preference in favor or entrepreneurial market entry. This would imply to align the tax rates that are applicable for the sale of a business or patent, and for the profits earned by the entrepreneur under market entry. These arguments support the policy recommendations in the Mirrlees Review (2011), which stress - in the context of UK tax policy - the case for more tax neutrality in business taxation. A final part of the appropriate policy package would be to improve the legal framework and to reduce the transaction costs for sales to incumbent firms, so as to ensure an effective bidding competition for target firms and their patents.

In addition to its normative implications, our analysis also leads to several hypotheses that can be tested empirically. As corporate tax rates for incorporated start-ups have fallen sharply in many countries, this has led to a rising tax advantage for market entry by entrepreneurs over the alternative of project sale. According to our analysis, these developments should have led to a rising share of innovations that are commercialized by the market entry of entrepreneurs and at the same time should have induced less risky, and also less efficient, innovation projects.

In concluding, we emphasize that our analysis is but a first step towards a more comprehensive study of the effects of public policies on the market for entrepreneurial innovations. A possible first extension of our framework is to endogenize optimal policies towards small businesses, including the extent of loss offset provisions that apply to new enterprises. In such a setting the disadvantages of restrictive loss offset opportunities that have been the focus of our study could be set against the possible tax revenue losses that arise from fraudulent claims for tax rebates. Another extension would be to introduce a more complex set of contractual arrangements between innovators, banks and venture capitalists, where the incentives for all agents may be distorted by asymmetric information. For instance, heavy reliance on debt financing can lead to excessive risk-taking by entrepreneurs, counteracting the distortion that arises from limited loss offset opportunities in the present paper. Finally, the analysis could be extended to a dynamic model, in order to capture the growth-promoting effect that is a core reason for the policy support to entrepreneurship.

Appendix: Proof of Lemma 1

First, note that $b_i \ge \max v_{il}$, $l = \{e, i\}$ is a weakly dominated strategy, since no incumbent will post a bid equal toor above its maximum valuation of obtaining the

invention and that firm e will accept a bid iff $b_i > v_e$.

Regime 1: Consider equilibrium candidate $b^* = (b_1^*, b_2^*, ..., no)$, where $b_j^* < v_e \ \forall j \in J$. It then directly follows that no firm has an incentive to deviate and thus, b^* is a Nash equilibrium.

Then, note that the entrepreneur will accept a bid iff $b_j \ge v_e$. But $b_j \ge v_e$ is a weakly dominated bid in these intervals, since $v_e > \max\{v_{ii}, v_{ie}\}$. Thus, the assets will not be sold in these intervals.

Regime 2: Consider equilibrium candidate $b^* = (b_1^*, b_2^*, ..., yes)$. Then, $b_w^* > v_e$ is not an equilibrium since firm w would then benefit from deviating to $b_w = v_e$. Further, $b_w^* < v_e$ is not an equilibrium, since the entrepreneur would then not accept any bid. If $b_w^* = v_e - \varepsilon$, then firm w has no incentive to deviate. By deviating to $b'_j \leq b_w^*$, firm j's payoff does not change $(j \neq w, e)$. By deviating to $b'_j > b_w^*$, firm j's payoff decreases since it must pay a price above its willingness to pay v_{ii} . Accordingly, firm j has no incentive to deviate. By deviating to n_o , the entrepreneur's payoff decreases since it foregoes a selling price above its valuation v_e . Accordingly, the entrepreneur has no incentive to deviate and thus, b^* is a Nash equilibrium.

Let $b = (b_1, ..., b_n, yes)$ be a Nash equilibrium. If $b_w \ge v_{ii}$, then firm w will have the incentive to deviate to $b' = b_w - \varepsilon$. If $b_w < v_{ii}$, the entrepreneur will have the incentive to deviate to no, which contradicts the assumption that b is a Nash equilibrium.

Let $b = (b_1, ..., b_n, no)$ be a Nash equilibrium. The entrepreneur will then say no iff $b_h \leq v_e$. But incumbent $j \neq d$ will have the incentive to deviate to $b' = v_e + \varepsilon$ in Stage 1, since $v_{ie} > v_e$. This contradicts the assumption that b is a Nash equilibrium.

Regime 3: Consider equilibrium candidate $\mathbf{b}^* = (b_1^*, b_2^*, ..., yes)$. Then, $b_w^* \ge v_{ii}$ is a weakly dominated strategy. Also $b_w^* < v_{ii} - \varepsilon$ is not an equilibrium since firm $j \ne w, e$ then benefits from deviating to $b_j = b_w^* + \varepsilon$, since it will then obtain the assets and pay a price lower than its valuation of obtaining them. If $b_w^* = v_{ii} - \varepsilon$, and $b_s^* \in [v_{ii} - \varepsilon, v_{ii} - 2\varepsilon]$, then no incumbent has an incentive to deviate. By deviating to no, the entrepreneur's payoff decreases, as she foregoes a selling price exceeding her entry valuation v_e . Accordingly, the entrepreneur has no incentive to deviate and thus, b^* is a Nash equilibrium.

Let $b = (b_1, ..., b_n, no)$ be a Nash equilibrium. The entrepreneur will then say no iff $b_h \leq v_e$. But incumbent $j \neq e$ will then have the incentive to deviate to $b' = v_e + \varepsilon$ in Stage 1, since $v_{ie} > v_e$. This contradicts the assumption that **b** is a Nash equilibrium.

Literature

- Acs, Z.J., Audretsch D.B. (2005). Entrepreneurship, innovation and technological change. *Foundations and Trends in Entrepreneurship 1*, 149-195.
- Altshuler, R., Auerbach, A.J. (1990). The significance of tax law asymmetries: An empirical investigation. *Quarterly Journal of Economics* 105, 61-86.
- Anton, J.J., Yao, D.A. (1994), Expropriation and inventions: Appropriable rents in the absence of property rights. *American Economic Review* 84, 190-209.
- Auerbach, A.J. (2007). Why have corporate tax revenues declined? Another look. CESifo Economic Studies 53, 153-171.
- Ayers, B.C., Lefanowicz, C.E., Robinson, J.E. (2003). Shareholder taxes in acquisition premiums: The effect of capital gains taxation. *Journal of Finance 58*, 2783-2801.
- Baumol, W.J. (2004). Entrepreneurial enterprises, large established firms and other components of the free-market growth machine. *Small Business Economics 23*, 9-21.
- Bhattacharya, S., Mookherjee, D. (1986). Portfolio choice in research and development. RAND Journal of Economics 17, 594-605.
- Bloom, N., Griffith, R., Van Reenen, J. (2002). Do R&D tax credits work? Evidence from a panel of countries 1979-1997. *Journal of Public Economics* 85, 1-31.
- Bond, S., Devereux, M. (2003). Generalised R-based and S-based taxes under uncertainty. *Journal of Public Economics* 87, 1291-1311.
- Cabral, L. (2003). R&D competition when firms choose variance. Journal of Economics & Management Strategy 12, 139-150.
- Cohen, W. (2010). Fifty years of empirical studies of innovative activity and performance. In: Hall, B.H. and Rosenberg, N. (eds), *Handbook of the Economics of Innovation, Vol. 1*, 129-213. North-Holland, Amsterdam.
- Cullen, J.B., Gordon, R.H. (2007). Taxes and entrepreneurial risk-taking: Theory and evidence for the U.S. *Journal of Public Economics* 91, 1479-1505.

- Da Rin, M., Di Giacomo, M., Sembenelli, A. (2010). Corporate taxation and the size of new firms: Evidence from Europe. Journal of the European Economic Association 8, 606-616.
- Da Rin, M., Di Giacomo, M., Sembenelli, A. (2011). Entrepreneurship, firm entry. and the taxation of corporate income: Evidence from Europe. *Journal of Public Economics 95*, 1048-1066.
- Devereux, M.P., Lockwood, B., Redoano, M. (2008). Do countries compete over corporate tax rates? *Journal of Public Economics 92*, 1210-1235.
- Djankov, S., Ganser, T., McLiesh, C., Ramalho, R., Shleifer, A. (2010). The effect of corporate taxes on investment and entrepreneurship. *American Economic Jour*nal: Macroeconomics 2, 31-64.
- Domar, E.D., Musgrave, R.A. (1944). Proportional income taxation and risk-taking. Quarterly Journal of Economics 58, 388-422.
- Edgerton, J. (2010). Investment incentives and corporate tax asymmetries. *Journal* of Public Economics 94, 936-952.
- EFI (Expert Commission for Research and Innovation in Germany) (2011). Report 2011. http://www.e-fi.de/fileadmin/Gutachten/EFI_2011_en_final.pdf (English version).
- Egger, P., Keuschnigg, C., Winner, H. (2009). Incorporation and taxation: Theory and firm-level evidence. CESifo Working Paper No. 2685, Munich.
- Ernst, C., Spengel, C. (2011). Taxation, R&D tax incentives and patent application in Europe. ZEW Discussion Paper No. 11-024, Mannheim.
- European Commission (2008). Think small first. A "Small Business Act" for Europe. Communication COM (2008) 394. Brussels. http://ec.europa.eu/enterprise/policies/sme/small-business-act/index en.htm
- Färnstrand Damsgaard, E., Norbäck, P.-J., Persson, L., Vasconcelos, H. (2010). When do independent innovators make the break-through inventions? Mimeo, Research Institute of Industrial Economics, Stockholm.

- Friebel, G., Giannetti, M. (2009). Fighting for talent: Risk-taking, corporate volatility and organisation change. *The Economic Journal 119*, 1344-1373.
- Fuest, C., Huber, B., Nielsen, S.B. (2002). Why is the corporate tax rate lower than the personal tax rate? The role of new firms. *Journal of Public Economics* 87, 157-174.
- Gentry, W.M., Hubbard, R.G. (2000). Tax policy and entrepreneurial entry. *American Economic Review 90*, 283-287.
- Gilbert, R. (2006). Looking for Mr. Schumpeter: Where are we in the competitioninnovation debate? In: Jaffe, A.B., Lerner, J. and Stern, S. (eds.), *Innovation Policy and the Economy*, Vol. 6. MIT Press, Cambridge and London, 159-215.
- Giuri, P., Mariani, M., Brusoni, S., Crespi, G., Francoz, D., Gambardella, A., Barcia-Fontes, W., Geuna, A., Gonzales, R., Harhoff, D., Hoisl, K., Le Bas, Ch., Luzzi, A., Magazzini, L., Nesta, L., Nomaler, Ö., Palomeras, N., Patel, P., Romanelli, M., Verspagen, B. (2007). Inventors and invention processes in Europe: Results from the PatVal-EU survey. *Research Policy 36*, 1107-1127.
- Gordon, R.H. (1998). Can high personal income taxes encourage entrepreneurial activity? *IMF Staff Papers* 45, 49-80.
- Häckner, J. (2000). A note on price and quantity competition in differentiated oligopolies. *Journal of Economic Theory* 93, 233-239.
- Henkel, J., Rønde, T., Wagner, M. (2010). And the winner is acquired: Entrepreneurship as a contest with acquisitions as the prize. CEPR Discussion Paper No. 8147.
- Henrekson, M., Johansson, D. (2010). Gazelles as job creators: a survey and interpretation of the evidence. *Small Business Economics* 35, 227-244.
- Henrekson, M., Sanandaji, T. (2011). Entrepreneurship and the theory of taxation. Small Business Economics 37, 167-185.
- Hvide, H.K. (2009). The quality of entrepreneurs. The Economic Journal 119, 1010– 1035.

- Keuschnigg, C., Nielsen, S.B. (2002). Tax policy, venture capital, and entrepreneurship. Journal of Public Economics 87, 175-203.
- Keuschnigg, C., Nielsen, S.B. (2004). Start-ups, venture capitalists, and the capital gains tax. *Journal of Public Economics* 88, 1011-1042.
- Landsman, W., Shackelford, D. (1995). The lock-in effect of capital gains taxes: Evidence from the RJR Nabisco leveraged buyout. National Tax Journal 48, 245-259.
- Lerner, J. (2010). Geography, venture capital, and public policy. Rappaport Institute and Taubman Center Policy Briefs. March 2010.
- Mankiw, N.G., Whinston, M.D. (1986). Free entry and social inefficiency. RAND Journal of Economics 17, 48-58.
- Mirrlees, J., Adam, S., Besley, T., Blundell, R., Bond, S., Chote, R., Gammie, M., Johnson, P., Myles, G., Poterba, J. (2011). Small Business Taxation. In: Tax by Design: The Mirrlees Review, 451-469. Oxford University Press.
- Norbäck, P.-J., Persson, L., Svensson, R. (2010). Creative destruction and productive preemption. CEPR Discussion Paper No. 8281, London.
- OECD (2010). SMEs, entrepreneurship and innovation. Paris.
- Reinganum, J. (1989). The timing of innovation: Research, development, and diffusion. In: Schmalensee, R. and Willig, R. (eds.) Handbook of Industrial Organization. Elsevier, Amsterdam, 849-908.
- Rosen, R. (1991). Research and development with asymmetric firm sizes. RAND Journal of Economics 22, 411-429.
- Salop, S. (1979). Monopolistic competition with outside goods. Bell Journal of Economics 10, 141-156.
- Serrano, C. (2010). The dynamics of the transfer and renewal of patents. *RAND* Journal of Economics 41, 686-708.
- Singh, N., Vives, X. (1984). Price and quantity competition in a differentiated duopoly. RAND Journal of Economics 15, 546-554.

Tillväxtanalys (2010). Follow up of the newly established enterprises in 2005 - three years after the start. Statistik 2010:2, Sweden. www.tillvaxtanalys.se/tua/export/sv/filer/statistik/nyforetagande/Uppfoeljning_3_xr_nystartade_foretag_2005.pdf



Figure 1



Figure 2





